Author's response to reviews

Title: 'Palliative care': a contradiction in terms? A qualitative study among cancer patients with a Turkish or Moroccan background, their relatives and care providers

Authors:

Fuusje M De Graaff (f.m.degraaff@uva.nl)
Anneke L Francke (a.francke@nivel.nl)
Maria ETC Van den Muijsenbergh (M.vandenMuijsenbergh@elg.umcn.nl)
Sjaak Van der Geest (S.vanderGeest@uva.nl)

Version: 3 Date: 14 January 2010

Author's response to reviews:

Dear Mr/Mrs Alam.

Amsterdam, 14 January 2010

Please find enclosed our revised manuscript: ‘Palliative care’: a contradiction in terms? A qualitative study among cancer patients with a Turkish or Moroccan background, their relatives and care providers.

We submitted our manuscript yesterday and we hope we are still in time to admit our changes before you sent it to your peer reviewers. The main mistake of the first draft is that Prof Sjaak van der Geest is not using his initials. Although his initials are JDM, as an author the name should be Sjaak van der Geest.

And the text has been checked for typing and grammatical mistakes, but the content had not been changed.

Yours faithfully,

Drs. Fuusje de Graaff,

also on behalf of the co-authors Prof. Dr. Anneke Francke, Dr. Maria van den Muijsenbergh, Prof.dr. Sjaak van der Geest